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Abstract

As the unidirectional-solidifying melt rotates with respect to an inclined axis, a shear flow is induced to move parallel
to the melt/solid interface while it changes its direction along the axis perpendicular to the interface, like an Ekman
spiral flow. This induced flow will in turn alter the morphology of the solidifying melt/solid interface and eventually
influence the quality of the final casting. In the present paper, we investigate this induced flow in the system rotating in a
general way while physical conditions vary: different rotating conditions, different thermal and solutal gradients, and
the non-Boussinesq effect due to variable viscosity of the melt. Results show that, in brief, the induced flow is driven by
the gravity due to inclination while modified by rotation. As each of the following parameters is higher: the effective
Taylor number (7¢), the thermal Rayleigh number (R;), the solutal Rayleigh number (R.), or the Lewis number (Le), the
induced flow is of larger velocity. The variable-viscosity effect is more complex, depending on the value of other physical
parameters. © 2001 Elsevier Science B.V. All rights reserved.
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1. Introduction meantime, a compositional boundary layer forms

above the interface because the solidifying binary

The directional solidification is one of the major
industrial techniques by which semiconductor
materials such as silicon and gallium—arsenic are
manufactured. During the solidification process,
the latent heat is released from the melt/solid
interface due to phase change and is conducted
away into the solid and the melt, forming a
thermal boundary layer above the interface. In the
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solution rejects or incorporates the solute at the
interface, enhancing the solutal diffusion in the
vicinity of the melt/solid interface. After the
thermal and compositional boundary layers form,
the morphological instability may occur and
disturb the interface into non-planar shape [1].
On the other hand, another kind of instability, or
the convective instability, can also occur if the
rejected solute is lighter or the incorporated solute
is heavier than the solvent, rendering the density
distribution in the compositional boundary layer
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hydro-statically unstable and eventually leading to
the onset of double-diffusive convection above the
interface [2]. This convective instability may
couple with the morphological instability and
result in an undesired change of the quality of
final casting. An effective scheme to control these
two instability modes has accordingly become an
important issue discussed in the research of
directional solidification, and meanwhile the im-
position of a shear flow in the melt of directionally
solidifying solution has been shown to be a
potentially effective scheme to control these
stability modes as well as to enhance the quality
of the final casting.

The influence of the imposed shear flow on these
two instability modes was first investigated by
Delves [3,4] who imposed quadratic and Blasius
boundary-layer flows and by Coriell et al. [5] who
imposed a plane Couette flow onto the melt/solid
interface. Later, MacFadden et al. [6] imposed a
plane stagnation flow vertically onto the melt/solid
interface and Forth and Wheeler [7,8] imposed an
asymptotic suction boundary-layer flow above the
interface. All these studies concluded that the
imposed shear flow inhibits the instability modes
propagating in a direction being non-perpendicu-
lar to the imposed flow (the modes containing
transverse components). Recently, however,
Chung and Chen [9] showed that as the imposed
shear flow (a Blasius type boundary-layer flow)
becomes strong enough, a transverse mode (roll-
axis perpendicular to the imposed flow) of very
unstable characteristics is induced. This mode
could exist even without buoyancy. Instead of
imposing the shear flow artificially as did by the
above-mentioned studies, Sample and Hellawell
[10] proposed to rotate the casting mode with
respect to an axis inclined to the gravity so that a
shear flow can be induced naturally due to
inclination. Although no vigorous evidence
was provided, they nevertheless conjectured that
the shear flow might inhibit greatly the instability
of the melt solidifying from below. Their con-
jecture was later investigated analytically by
Chung and Chen [11], in which they showed that
a strong helical shear flow was induced and the
shear flow generally enhanced the stability of the
system.

Since the inclined rotation generates a naturally
induced shear flow, instead of an artificial imposed
flow as did by previous studies, it is therefore more
feasible for the industrial purpose. We accordingly
investigate in the present paper in a broad physical
sense to study the change of the induced flow
under various physical situations. Special empha-
sis is placed on both the effects due to varying
physical parameters and the non-Boussinesq effect
due to temperature-dependent viscosity because
the viscosity of the melt may change significantly
during solidification due to temperature variation.
For example, the viscosity of glass forming alloys
decreases by 0(10%°) when temperature decreases
from working condition to melting point [12] and,
for lead—tin alloy, it changes by a factor of two
when temperature changes over 200 K [13]. More-
over, since the temperature-dependent viscosity
changes dramatically in the region above the melt/
solid interface because of the sharp temperature
variation, it is expected that the temperature-
dependent viscosity may also influence signifi-
cantly on the induced shear flow, which is largely
confined to the region above the interface.

In the present paper, we consider the directional
solidification of binary alloy under inclined rota-
tion, in which a shear flow is possibly induced by
inclination and modified by rotation as well as the
viscosity variation. We present the mathematical
model of the flow induced by the inclined rotation
in Section 2. The equations are solved by an
analytical approach and the solutions are shown in
Section 3. Based on this solution, the features of
the induced flow are discussed in Section 4. The
changes of the induced flow under different
physical situations including variable viscosity
are discussed in Section 5. Finally, concluding
remarks are given in Section 6.

2. Problem description and formulation

Consider the system as shown in Fig. 1, a dilute
binary solution of temperature 7',, and concentra-
tion C,, is solidifying upwards, in which a solid
layer is formed below the semi-infinite bulk melt.
The melt/solid interface described by z = h(x, y, 1)
is assumed to be initially planar and advances into
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Fig. 1. The schematic description of the system considered. The solid layer is solidifying with a constant velocity V' upwards. An
induced flow moves parallell along the melt/solid interface. The casting mode rotates, including spinning and precession, with respect

to an inclined axis.

the fluid with a constant speed V. The cooling tank
rotates in a general way including precession and
spin that the angular velocity can be expressed by

® = (¢, sin ¢, sin ¢ )e, + (¢, sin ¢, cos P,)e,
+ (hp €08 by + e, (R))

where ¢,,, ¢, and ¢, are the angles of precession,
nutation and spin, respectively and ¢, and ¢, are
the angular velocities of precession and spin,
respectively (see Fig. 1). Besides, e,, e, and e. are
the unit vectors of Cartesian coordinate, which sits
on the melt/solid interface as denoted by the
x—y—z frame in Fig. 1.

In such a coordinate system, the governing
equations in the fluid region h<z< oo are

V-u=0, (2.2a)
0 0 .
(a_t_ V§+U-V)u+2¢><u

__vr, (ﬂ— 1>g+piv-(un>, (2.2b)
0

Po Po
o i+u~v C = DiV*C (2.2¢)
o oz T '
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(Q — Vg +u- v) T = kiV2T. (2.2d)

ot 0z

In above equations, u represents the velocity
vector (u,v,w) measured with respect to the
cooling tank, P is the pressure, and p, is the
reference density. The density of the fluid is
assumed to be constant except in the gravity term
that p = po(1 — (T — T) — BC), where « and f
are the thermal and solute expansion coefficients,
respectively, and T, represents the freezing
temperature of the pure solvent. The vector of
gravitational acceleration g = —g(sin ¢, sin ¢pe, +
sin ¢, cos ¢.e, + cos ¢ e.) depends on both the
nutation and spin angles, where ¢ is the gravita-
tional constant. D = Vu+ Vu' is the deviatoric
strain tensor and the superscript “T”’ denotes the
transverse of the tensor, C is the concentration, Dy
is the solute diffusivity, T is the temperature and k¢
is the thermal diffusivity. The viscosity of the fluid
w is considered as an exponential function of
temperature

1= po exp[—c(T — Tin)], (2.3)

where ¢ >0 is an arbitrary constant, indicating
that the viscosity of the fluid decreases with
increasing temperature, and pu, = vop, is the
reference dynamic viscosity, vy is the reference
kinematic viscosity. The exponential model is
applied because of its wider use for hydrogen-
bonded liquids than the other models [14], which
suits the aqueous ammonium chloride solution
considered in the present paper well.

In the solid region z <A, we neglect the diffusion
of solute concentration and consider the diffusion
of heat only. Then the heat equation is

0 0 )

(61 Vaz> T=kV-T, 2.4
where ks is the thermal diffusivity of the solid
phase. The boundary conditions at infinite far field
are assumed to be that the fluid experiences a rigid-
body rotation with the tank and both the
concentration and temperature of the fluid remain
unchanged during solidification process. Further-
more, it is assumed that the height of the tank is
large enough that the influence on the induced flow
due to the possible deformation of the free surface

can be neglected. Accordingly, at the infinite far
field z— oo we have

u—»0, C-C,, T-T,. (2.5)

At the melt/solid interface z = h(x, y, t), the
boundary conditions are

uxn =0, (2.6a)

u-n=0, (2.6b)
+

Cta =k V+ Oh e.-n= —Df—ac , (2.6¢)
ot On

TH =mCt + Tn(l — ), (2.6d)

TH=1T", (2.6e)

oh oT~ oT™
L(V+§>ez ‘n = KsE* KFE’ (2.6f)

where n is the unit vector normal to the interface
directing toward the fluid. The superscripts ““+
and “—" account for the quantities right above
and below the interface, respectively. Eq. (2.6a) is
the no-slip condition. Eq. (2.6b) expresses the
conservation of mass at the interface by neglecting
the difference of the densities between the solid
and liquid phases. Eq. (2.6¢c) accounts for the
conservation of the solute across the interface,
where k = C~/C* is the segregation or partition
coefficient. Eq. (2.6d) is the thermal-dynamical
equilibrium condition that describes the depen-
dence of the freezing temperature of a binary alloy
upon its composition. The capillary effect (the
Gibbs—-Thompson effect) is included in the last
term and the liquidus slope m is assumed to be a
constant. I' is the capillary length and ¢ is the
curvature of the interface that is assumed to be
negative for a concave projection into the fluid.
Eq. (2.6e) expresses the continuity of the tempera-
ture across the interface and Eq. (2.6f) accounts
for the energy balance at the interface, where
and xy are the thermal conductivity of the solid
and fluid, respectively, L is the latent heat per unit
volume of solid.

The governing equations as well as the bound-
ary conditions are non-dimensionalized with
the following solute-field scales: V' for velocity,
H = D;/V for length, D;/V? for time, C, for
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concentration, Ty, for temperature and vop, & /Ds
for pressure. Note that the temperature is sub-
tracted by Ty, before dividing it by the scale. Then
the dimensionless viscosity function p can be
written as

p=exp[y(=T/Tx)] (2.7)

where y = In(y/u,,) accounts for the viscosity
contrast between the fluid at the freezing tempera-
ture and the bulk fluid at infinity, 7 is the
dimensionless temperature and 7', is the dimen-
sionless temperature at infinity. As a result,
the dimensionless governing equations in the
fluid are

V-u=0, (2.8a)

oP
=~ 5 T (RC+ RIDISaS,(1)

+ (V- uD) - ey
n {(—1)”"Ta1])/2Cn n (—1)"5Ta;/2] v

- [(—1)”vTa;/ZSncs(z)} W, (2.8b)

oP
:_a—+(R C+ RIS\ Cs(t) + (V- uD) - e,

+ [ 1mTay?e, + (-1 Tal

+ [(— 1)"PTa]13/zSnSs(t)} W, (2.8¢)

o 0
71 ~
Sc (at az+u V)w
oP
:—a—+(RC+RtT)C

+ (V- D) - €. + | (=1)"*Tal/25,C(0)|u

_ {(71)”PTar‘)/2SnSs(t)} v, (2.8d)

3 ,
9_9 - 2.
(az S tu v)c v2C, (2.8¢)

Le”! (g — % +u- v) T =VT, (2.8f)

and the dimensionless heat balance equation in the
solid layer is
o 0

Ls ' (=—=— )T =V’T. 2.9

g (ar az> 29)
In above equation, Sc = vy/Ds is the Schmidt
number, Le = k¢/Dy is the Lewis number of fluid
and Ls~' = ky/Dy is the Lewis number of solid, R.
and R; are the solute and thermal Rayleigh
numbers, respectively defined by

9BCo H? go TwH?

Re="—"— R ="—. 2.10

C Vo Df ) t Vo Df ( )

The parameters Ta, and Ta, are the Taylor
numbers of precession and spin, respectively, and
are defined by

. 2 .2
2H? 2
Ta, = ( ¢P> . Ta, = <2H ¢5) . @1
Vo Vo

Some abbreviations have been used and their
definitions are shown in the following:

Sy =sin¢,, C,=cos¢,, Ss()=sin(Qr),
Cs(1) = cos (Q1), (2.12)

where Q = Sc(—l)"sTa;/ 2/2 is the dimensionless
angular velocity of spin. Note that the exponents
n, and 7, indicate the sense of precession and spin,
respectively: As (,bp >0 or ¢, >0, one obtains np =
0 or ng =0, respectively; as qu <0 or d)s <0, it
yields n, = 1 or ng = 1, respectively.

The dimensionless boundary conditions at
z— 00 are

u=0, C=1, T=T,, (2.13)
at the melt/solid interface z = h(x, y, f) are

uxn =0, (2.14a)
u-n=0, (2.14b)

+ Oh +
C(k—=1) 1+a e.-n=V(_C" -n, (2.14¢)
TH=MC" —yé, (2.14d)
TH=T", (2.14¢)
Oh I +

L1+ )& n=(Ls VT~ —LeVT")-n,

(2.14f)
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where the parameter M = mC, /Ty, is the dimen-
sionless liquidus slope, # = I'V /Dy is the dimen-
sionless capillary length and & = L/Tn(pCp)~ is
the Stephan number. Note that in Eq. (2.14f) the
difference of the specific heat per unit volume
between the solid and liquid phase is neglected.

3. Analytical solution of the flow induced by
inclination

The induced flow during solidification before
the melt/solid interface losing the planar shape is
induced by the inclination of cooling tank while is
modified by rotation. To investigate this induced
flow analytically, it is assumed that the melt/solid
interface remains planar during solidification,
namely,

hy = 0. 3.1

It is assumed also that the horizontal dimension
of the cooling tank is much larger than the
characteristic length of the solute scale H. Conse-
quently, it can be seen from the scale analysis of
the continuity equation that the velocity along the
height of the tank is much weaker than that along
the x—y plane when small-scaled convection is
absent from the system, and the differential
derivatives in the x—y planes are negligible in
comparison with that along the height of tank as
the horizontal size of tank is sufficiently large. As a
result, the temperature and concentration are not
affected by the presence of the flow induced by
inclination and then should be similar to those of
previous studies [6,8,15], which are shown in the
following for convenience for subsequent discussion.

In the melt z>0, the concentration and
temperature distributions are represented by a
concentration boundary layer and a thermal
boundary layer, respectively, i.e.,

Cp=1-Ge 7, (3.2)

T, = T, — LeGie /!¢, (3.3)

where the local concentration gradient and tem-
perature gradient are
k—1

G, = % 3.4

Ty —M/k
Le
In the solid layer z<0, the concentration

keeps constant and the temperature decreases
exponentially,

C, =1, (3.6)

G, = (3.5

Ty =T, + L — (LLeG)e /5, (3.7)

Regarding the induced flow, it is shown from
scale analysis that the z-component of the induced
velocity can be assumed to be zero, i.e.

wy = 0. (3.3)

By applying Egs. (2.8b)—(2.8d) and boundary
condition Equation (2.13), we obtain the pressure
distribution in the melt as

P, = Pb(Z, B+ (Re + R Too)[xSnSs(?)

+ySn Cs([) + ZCn] (39)

The reduced pressure Py(z, ) can be obtained by
substituting Eq. (3.9) into Eq. (2.8d) yields

al_)b n 1/2

=== DTS 0|

= [ Tays, 8,0 o

+[Re(Co — D) + R(To — T)]Cn.

Note that the pressure gradient in z-direction
balances the z-component of the fluid weight,
leading the melt far above the solidifying front
to be quiescent and makes the whole melt
motionless in the z-direction. Namely, in the far
field from the interface, the pressure gradient
can be seen as the buoyancy due to the
density variation in the z-direction. On the
other hand, however, due to the fact that the
density changes greatly within both of concentra-
tion and thermal boundary layers, the pressure
gradient in both x- and y-directions cannot
balance the weight of the fluid along correspond-
ing directions and causes a flow along the inter-
face, which is then modified by the Coriolis force
due to the rotation. The x- and y-components u,
and vy, of the induced flow can be solved by
substituting Egs. (3.8) and (3.10) into Egs. (2.8b)

(3.10)
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and (2.8c¢), yielding
d*Uy,
a2
=[Re(Co — D) + R(To — T0)]Sh, (3.11)

dU .
+(,u'+Sc")d—be iT. Uy

where Up = up + ivp = Ubei‘f’g,qﬁg = —(Qt+7n/2),
W =du/dz and T, is defined as

_1 s
Te = (-1)"Ta}*C, +%Ta;/2. (3.12)

The associated boundary conditions are

U,=0, atz=0 and z— 0. (3.13)
Since the gravity component on the x—y plane
can be expressed by g, + ig, = Syge's, where ¢,
indicates the orientation of the gravity component
in the x—y plane. Egs. (3.11) and (3.13) can be
solved by asymptotically expanding the dependent
variables with respect to ¢, where ¢ =1/Le is
usually a small parameter for an alloy,

Up = Upo + eUpy + &2 Upa + -+, (3.14)
and the viscosity p can be expressed as

u = Aexp[Bexp(—ez)]
B+ B
:Q{l—sBz—ksz(——; )22

B+ 3B
—83( +63 >23+-~}, (3.15)

where A =exp(—y), B=17LeG|/T,, and Q=
Aexp B. By substituting Egs. (3.14) and (3.15)
into Egs. (3.11) and (3.13) and by equating the
same order terms, we obtain the following
perturbation equations of different order of ¢

,dUno
c

d*0 .
0(?0) . Q b0 +S — iTeUbO
dz2 z
= [Re(Co — 1) + R(Tv — T)ISh, (3.16a)
d*0 du .
1y . bl _1 dUpi
OE): 0 i2 Sc . iT Uy
d*Uo  dUy
= QB|:Z i2 + & }, (3.16b)

d*Up, _,dUy,

2 . TN
oE): 0 a2 + Sc a- iT.Uyy
d*U,,  dUy,
= 0Bz dz2 + dz
(1+ B) ,d* U d U
— —-(1+B 1
5 z 02 (1+ B)z i (3.16¢)

The terms with higher order than O(e?) are
neglected. The perturbed boundary conditions are
Up=Uy =Up=0 atz=0 and z— 0.
(3.17)
The solutions of Egs. (3.16) U, U, and Uy, can

be obtained by a straight forward derivation,
yielding

Upo = Aje ™" 4 Are™ 7™ — (4, + A2)e®,  (3.18a)
Ubl = (BlZ + B2)372 + (B3Z + B;;)eiz/Le

+ (Bsz*> + Bz — By — By)e”, (3.18b)
sz = (C122 + Crz+ C3)e™*

+ (C422 + C5Z + Cﬁ)e_z/Le

+ (C7Z4 + CgZ3 + C922

+ Cioz — C3 — Cg)e“, (3.18¢c)

where the definitions of the coefficients A and 4,,
B1—B¢ and C|—Cj( are shown in the Appendix A,
and

a=—d;' —if, (3.19a)

dg = [20Se) ™! + ¢!, (3.19b)

o — {(QSC)_2 +[(0Se)™* + 16(Te/Q)2]1/2}1/2
= - ,

(3.19¢)
f =

; {_(Qsc)z 1080+ 16(T6/Q)2]'/2}1/2
|Te| 8 :

(3.19d)

The combined effect due to precession and spin on
the amplitude of induced velocity are accounted
for by the parameter 7, of Eq. (3.12).
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4. The induced flow

From Eq.(3.18) one can see, whether the
viscosity is constant or variable, the direction of
the induced flow varies with time periodically with
a frequency equal to the spin angular velocity Q.
The induced velocity varies with the height of the
tank and can be separated into three components
of different length scales. The first component is
of the length scale of the solute boundary
layer, varying vertically with the exponential

function e * and will be called the solutal-layer
flow. The second component is of the length
scale of the thermal boundary layer, varying
with e %/ and will be called the thermal-layer
flow. The third component varies its magnitude
with e %/ and changes its direction with a period
2n/f, being apparently a kind of spiral Ekman
flow and will accordingly be called the Ekman-
layer flow.

To illustrate more clearly the variation of these
three components, we show in Fig. 2 the variations

1] e e —— -
- (a) - ]
I
8 - -
|
L | J
|
6 - -
|
z : B : b
4 K Thermali B }Solutc _
1 |
t Ekman ! P | E
\
2k | : : N
" Solute i Thermal}
\ i i
0 = 1 1 i M IR B
0 20 40 60 80 0 90 180 270 360
U, /S, ¢~ b, (deg.)

(©

10

30 —100

Fig. 2. An example of the induced flow. (a) The velocity amplitudes along the z-axis of the three components: the solutal layer flow, the
thermal layer flow, and the Ekman layer flow. (b) The phase angle variations of the three flow components; the phase of Ekman layer
flow changes periodically with height. (c) The three-dimensional velocity profile of the induced flow, showing that the spiral structure is

mainly confined to a shallow layer above the melt/mush interface.
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of both the velocity magnitude and the direction
along with the tank height for the case T, =1,
R. =10, R, = 250, Le = 3600, Sc = 81, which cor-
responds to the aqueous ammonium chloride
solution of constant viscosity (i.e. y = 0) under
small angular velocity rotation (such as 1rpm)
with an inclined angle about 10°. It is seen from
Fig. 2(a) that both the solutal-layer flow and the
Ekman-layer flow are of the maximum velocity at
the melt/solid interface and decay exponentially
with height, while the thermal layer flow is
essentially of constant velocity along the height
and is of largest velocity compared to the other
two components. By combining these three com-
ponents, the induced flow vanishes at the interface,
satisfying the boundary condition, while increases
its velocity exponentially with height within the
region where both the solutal-layer flow and
Ekman-layer flow decay upwards. After reaching
the maximum, the induced flow velocity reduces a
bit and remains to be that of thermal-layer flow up
to the far field. In other words, the induced flow is
dominated by the thermal-layer flow in a large
region of the melt, while both the solutal-layer
flow and the Ekman-layer flow contribute to the
induced flow by reducing the thermal-layer flow
exponentially to zero when across the shallow
region above the interface.

On the other hand, the three component-flows
are of different directions, which can be seen from
Fig. 2(b) where ¢, is the phase angle of Uy. The
solutal-layer flow and the thermal-layer flow are of
constant directions, differing from the direction of
gravity by 45° and 90°, respectively, while the
direction of Ekman-layer flow changes periodically
with height, forming the main body of spiral
structure. To illustrate the spiral structure, we
show the velocity vector in Fig. 2(c) that the
induced flow in the far field stays in the y—z plane,
while near the interface the flow has a component
in x-direction. This component has its maximum
on the interface and decays rapidly with height.
Note that the direction of thermal-layer flow on
the interface is virtually opposite to that of
Ekman-layer flow, the combination of these two
component-flows therefore becomes a flow of
small velocity, which is then vanished due to the
solutal-layer flow, leading to the non-slip condi-

tion on the interface. At zx 3.2, on the other hand,
the Ekman-layer flow and the thermal-layer flow
are of the same direction, resulting in the largest
amplitude of the induced flow velocity. A small
contribution is made by the solutal-layer flow,
which modified a little the position of maximum
velocity.

The coefficients of the induced flow velocity of
Eq. (3.18) contain the parameter S,, indicating
that the induced flow is a result of the imbalance
between gravity and pressure gradient along x—y
plane when the angle between these two forces is
larger than zero. On the other hand, one may infer
that as the system is rotating with respect to the
vertical axis, i.e. the gravity and the pressure
gradient are in the same direction, they will have
no induced flow irrespective of the fluid viscosity
being constant or variable. Because of the inclined
rotation, the fluid is driven by the resultant force
of gravity and pressure gradient to move parallel
along the x—y plane. When the rotation includes
spin, the induced flow varies periodically with a
frequency (2, the spin angular velocity, because
both the gravity and pressure gradient change their
directions with Q. While this case cannot be
applied to the case of precession, since the phase
angle difference between the induced flow and the
gravity is independent of time, implying that the
induced flow due to precession shall be a steady
flow. This physical explanation can also apply to
the occurrence of the three component-flows: Both
the thermal-layer flow and the solutal-layer flow
are the results of system inclination while the
Ekman-layer flow has mainly resulted from system
rotation. In brief, one may conclude that the
induced flow is driven by system inclination and is
modified by the system rotation.

5. The physical parameter effects

In the present system there are several para-
meters affecting the structure of induced flow.
They are the effective Taylor number T., the
solutal Rayleigh number R, the thermal Rayleigh
number Ry, and the Lewis number Le. We examine
the effects of these parameters while consider the
fluid having constant viscosity, i.e. y = 0. On the
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other hand, we also consider the variable viscosity
effect on the induced flow, which turns out to be
crucial and complicated. The solution of the
induced flow is shown in Egs. (3.18) and (3.19),
based on which the following discussion is made.

5.1. Effective Taylor number effect

The induced flow is affected by three physical
parameters accounting for the inclined rotation:
the inclined angle, the rotation speed, and the
sense of rotation. As indicated by Chung and Chen
[11], a larger inclined angle leads to a larger
induced flow, a higher rotation speed induces a
flow of smaller velocity, and the senses of spin and
precession modify the induced flow. The effective
Taylor number defined in Eq. (3.12) is able to
account for the combined effect of these three
physical parameters. According to Eq. (3.12), a
larger 7T, does not necessarily imply either larger
spin speed or larger precession speed, depending
on the relative sense between spin and precession
and the inclination angle (accounted for by C,) as
well. For example, as both spin and precession are
of the same sense, i.e. n, =n, =0, a larger Tt
means a smaller inclined angle and/or higher spin

speed (accounted for by Tag) and precession speed
(by Tap). As spin and precession are of different
sense of rotation, for example n, = 1, ny = 0, the
relation among T, Cy, Ta, and Ta; is rather mixed,
no general trend can be followed with. We shall
therefore consider the first case, i.e. n, = n; = 0, to
interpret the results regarding the effect of Te..

A typical result is shown in Fig. 3(a), showing
that the induced flow velocity decreases as T,
increases, or equivalently the induced flow is
inhibited by a higher rotation speed through the
action of Coriolis force. In other words, because
the Coriolis force balances with the induced flow
and rotation, the induced velocity must accord-
ingly decrease as the rotation speed increases
(larger T.). On the other hand, a larger inclination
angle (a smaller C,, and a smaller T¢) leads to a
larger induced flow because the buoyancy compo-
nent along the melt/solid interface increases. We
note that the induced flow is also seen in the case
as T.—0, implying clearly that the flow is induced
by inclination, not by rotation. The T, effect on
the direction of the induced flow is constrained in
the shallow layer above the interface (Fig. 3(b))
because its influence is significant only on the
Ekman-layer flow, while is negligible on the

U/S:

100 250 270 290 310 330

¢, — @, (deg)

Fig. 3. The effective Taylor number (7¢) effect on the induced flow. A higher T, leads to an induced flow of smaller velocity amplitude.

(a) The velocity amplitude (b) the phase angle.
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Fig. 4. The effect due to solutal Rayleigh number (R.) variation, which contributes more significantly on the change of phase angle

than the amplitude. (a) The velocity amplitude (b) the phase angle.

solutal-layer flow and is of virtually no influence
on the thermal-layer flow.

5.2. Solutal Rayleigh number effect

The effect of solutal Rayleigh number is shown
in Fig. 4 for the case T, =1, R =250 and Le =
3600. It is known that solutal Rayleigh number
generally accounts for the density gradient due to
solute variation in z-direction and thus the buoy-
ancy to drive convection. Because of the solutal
gradient, a change of R, will influence both the
solutal-layer and the Ekman-layer flows because of
the interaction between Coriolis force and the
buoyancy. As a result, the influence of R. is
confined within the region in which the spiral-flow
structure prevails while it has virtually no influence
on the thermal-layer flow which dominates the
induced flow beyond the spiral-region. The results
of Fig. 4 show that, as R is small (=10-100), its
effect on either the solutal-layer flow or the
Ekman-layer flow is negligible so that the induced
flow is dominated by the thermal-layer flow. As R,
is large enough, its effect on both solutal-layer flow
and Ekman-layer flow prevails, resulting in a
dramatic change of velocity profile (in both
amplitude and direction) within the spiral-region.

By and large, the induced flow increases dramati-
cally the velocity within the shallow region above
the interface and then decreases dramatically
within another shallow region, and finally in-
creases again to approach the thermal-layer flow in
the outer region of the melt. On the other hand, an
increase of R. leads to an increase of velocity
gradient above the interface, which may in turn
influence the stability of morphology of the
interface. It is therefore an important factor for
precise control during the solidification process.

5.3. Thermal Rayleigh number effect

The thermal Rayleigh number R; accounts for
the density gradient due to temperature variation,
which is a stabilizing factor for convective flow
while it is a driving force to the induced flow when
interacting with the Coriolis force. Accordingly, a
change of R; influences both the thermal-layer and
Ekman-layer flows, as shown in Fig. 5 in which the
case T. = 1, R. = 10 and Le = 3600 is considered.
As R, increases, both the thermal-layer and
Ekman-layer flows increase their velocity ampli-
tude, and as a result, the induced flow becomes
more intensive (Fig. 5(a)). On the other hand, R,
has insignificant influence on the direction of
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Fig. 5. The effect due to thermal Rayleigh number (R;) variation, which contributes mainly on the velocity amplitude while has no
influence on the phase angle. In general, a larger R, results in a larger induced flow. (a) The velocity amplitude (b) the phase angle.

induced flow (Fig. 5(b)), because most of the R;-
effect is reflected through the change of thermal-
layer flow, which does not change the direction in
the melt.

5.4. Lewis number effect

Lewis number Le accounts for the ratio of
thermal diffusivity to solutal diffusivity, or equiva-
lently for the relative thickness of thermal bound-
ary layer to solutal boundary layer above the
interface. Namely, a larger Le means a thicker
thermal boundary layer above the interface in
comparison with the solutal boundary layer.
Accordingly, as Le increases, the thermal gradient
interacts with the Coriolis force in a thicker layer
above the interface, resulting in a more intense
induced flow (Fig. 6(a)). As Le decreases, on the
other hand, the induced flow is less intense and the
change of direction of the flow within the spiral-
region is more obvious (Fig. 6(b)) because the
solutal-gradient is becoming active in a thicker
solutal boundary layer.

5.5. The induced flow of variable viscosity

To investigate the variable-viscosity effect on the
induced flow, we consider the viscosity variation

ranging from y = 0 to 10, which accounts for an
e'%-fold (or a more than 20000-fold) change of
viscosity due to temperature variation. The analy-
tical solution of the induced flow is shown in
Egs. (3.18) and (3.19) and some typical results are
shown in Fig. 7 for the case R, =250 and 7. = 1
when R., Le and y vary. Results of Fig. 7 show
that the influence of variable-viscosity, accounted
for by changing y, varies for different R, and Le.
As Le is small, see Figs. 7(a), (d) and (g), a small
R. (=10) leads to a larger viscosity in a larger
region above the interface, resulting in that the
induced flow is retarded by the large viscosity in a
larger region. As R, increases, the induced flow is
confined to the shallower region above the inter-
face because of the increasing prevailing of the
interaction between buoyancy and Coriolis force
within this region. The induced flow is virtually
vanished when y is large, implying that the
importance of viscosity-variation predominates
when R, is large.

As Le becomes large, the importance of
viscosity-variation accounted for by varying y
decreases, as shown by Figs. 7(b), (¢) and (h). This
is obviously due to the fact that the change of
viscosity in this case has little effect on the induced
flow, which can be seen also from Figs. 7(c), (f)
and (i) where the effect of viscosity variation is
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Fig. 6. The effect due to Lewis number (Le) variation. A larger Le leads to a larger induced flow. (a) The velocity amplitude (b) the

phase angle.

very small. A larger Le implies a thicker thermal
boundary layer, so that the viscosity contrast
across the boundary layer is smaller than the case
of smaller Le provided that the temperature
difference across thermal boundary layer is the
same. Consequently, as Le is large enough (such as
3600), the viscosity contrast across thermal
boundary layer is relatively small, and therefore
the influence due to temperature-dependent visc-
osity on the induced flow is also small.

6. Concluding remarks

We have investigated analytically the flow in the
directional-solidifying melt under inclined rotation
including spin and precession. Results show that
the flow is driven by the gravity component along
the melt/solid interface and is modified by the
rotation including spin and precession through the
interaction between the Coriolis force and the
density gradient. The induced flow moves parallel
to the interface while it changes the direction along
the direction perpendicular to the interface, form-
ing a spiral structure like an Ekman flow. Neither
of the temperature nor the concentration distribu-
tion is influenced by the formation of induced
flow. The induced flow consists of three compo-

nents: the solutal layer flow, the thermal layer
flow, and the Ekman Ilayer flow; the third
component dominates the induced flow by chan-
ging the flow direction periodically along vertical
axis. The induced flow is influenced by the
following physical parameters:

(a) The effective Taylor number (7¢): The induced
flow increases its velocity when inclined angle
is larger, decreases when rotation speed is
larger, and is modified by the senses of spin
and precession. The effective Taylor number
defined in Eq. (3.12) is able to account for the
combined effect of these three physical factors,
leading to the fact that a smaller 7, results in a
more intensive induced flow.

(b) The solutal Rayleigh number (R.): an increase
of R, leads to an increase of velocity gradient
above the melt/solid interface, which may in
turn change the morphology of the interface
and the quality of the final casting. It is
therefore an important factor to the manu-
facture of high quality casting via a precise
control on the solidification process.

(¢) The thermal Rayleigh number (R;): As R;
increases, the induced flow becomes more
intensive because, through the interaction
with the Coriolis force, the thermal gradient
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Fig. 7. The effect of viscosity variation for different R, and Le. (a) R, = 10, Le = 300. (b) R, = 10, Le = 1000. (c) R. = 10, Le = 3600.
(d) R, =100, Le = 300. (¢) R, = 100, Le = 1000. (f) R, = 100, Le = 3600. (g) R. = 500, Le = 300. (h) R. = 500, Le = 1000. (i) R, =
500, Le = 3600.

(d)

is a driving force to the formation of induced
flow.

The Lewis number (Le): A larger Le means a
thicker thermal boundary layer above the
interface in comparison with the solutal
boundary layer. Accordingly, as Le increases,
the thermal gradient interacts with the Cor-
iolis force in a thicker layer above the inter-
face, resulting in a more intense induced flow.

(e) The temperature-dependent viscosity: The

effect due to viscosity variation is smaller
when either R. or Le becomes larger. This is
because, in general, the thickness of the
thermal boundary layer plays a crucial role
in determining the intensity of induced flow: a
larger viscosity contrast across the thermal
boundary layer leads to a more significant
influence on the induced flow.
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Appendix A

The coefficients of Egs. (3.18a)—(3.18c) are
shown as follows:

_ SaReGe g

A = : (A.1)
SnRtLeG] :
Ay = — T i A2
2 AL e, ( )
B, = %eid&’ (A.3)
Ac
B = G- VSIB s (A.4)
Ac
0BA; 4
= i A5
’ Le?A, ¢ (A.5)
20/Le — 1/Sc)B; .
B, = 2Q/Le—1/SA9Bs iy | p.. (A.6)
At
Bs = —QB(A; + Ay)a*/(4Qa + 2/Sc), (A.7)

Bs = —[QB(A4 + Ax)a + 20B5]/(2aQ + 1/Sc),

(A8)
Cl _ QB[Bl _(1 +B)Al/2]ei¢c’ (A9)
Ac
C, =
OB[B, — 3B1 + (1 + B)4,] + (40 — 2/Sc)C, it
Ac ’
(A.10)
Cs = OB(B, — By) + (2£i— 1/Sc)C; —20C, it
c
(A.11)
Cy = QB[B3 - (1 + B)A2/2] eiqil’ (A12)

LeAs

Cs =
OB[B;/Le? — 3B3/Le + (1 + B)Ay/Le] + (4Q/Le — 2/S¢)Cy
At
X e/, (A.13)
Co =
OB(B; — B4/LC) —20C4+ (2Q/Le — I/SC)C5
At
x ', (A.14)

C7 = QBa*Bs/(8Qa + 4/Sc), (A.15)
Cs = {OB[a*Bg + 5aBs

+ (1 + B)(A; + Ay)d*/2] — 120C7}/

(6Qa + 3/Sc) (A.16)
Cy = {QB[3aBs + 4Bs — (B, + By)d*

+ (1 + B)(A4:1 + Az)a]l — 60Cg}/

(4Qa + 2/Sc), (A.17)
Cio = {QB[Bs — (B2 + By)a] — 20Co} /

(2Qa + 1/Sc), (A.18)
where
Ac=[(Q — 1/Sc)* + TH'/,

_0—-1/Sc
e VI
. T.
sin d)c = XC’
Ar=[(QLe* - Sc™'Le”')* + 1717,
Le 2 — (ScLe)™!
cosqSL:[Q e AE cLe) ],
. T.

sin ¢, = AL (A.19)
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